I2P Guide: C. Andy Tsao Notes and Homework

Week 2. Fall 2002.

Outline

1. E 1_A = P (A). The fair price of a game (stake=1) = the probability of winning the game.

Dutch Book Argument. 2. Jargons Experiment The virtuality where the concerned process has unknown/uncertain outcomes. (The setting in which a ”probability” is considered) Event A subset of sample space on which the probability can be evaluated Sample Space The collection of all possible outcomes of an experiment 3. Examples: Coin-tossing experiment (one coin, two coins). Dice experiment, Course Grade ”experiment”.

4. Useful rules for calculating probabilities of events (Summary, p 17). Top-down, Bottom- up. 5. Building Blocks: Axioms of probability. Do you think they are reasonable? 6. Finite Sample Space

Homework (due 021004)

1. §1.4: 4, 7, 9. §1.5: 4, 7, 11. §1.6: 1, 4, 8.

Learning Guide and Announcement

1. Skim through Chapter 1–2, Intro and Summary. 2. Review §1.1–§1.6. Preview §1.7–§1.11.

3. There will be other assignments for coming weekw. Please do at least one problems per day. On P781–P799 of the textbook, there are answers to selected exercises.

4. Read §1.11. Good tips to learn along the way and useful in dinner party scenario. 5. The TA’s and their office hours will be announced shortly.

6. Typo correction: http://www.stat.cmu.edu/∼mark/degroot/typos/typos.html ♠♥♦♣

Page 1 of 1 Typeset by AMS-LaTeX